REGULAR PAPER

# Defining and evaluating network communities based on ground-truth

Jaewon Yang • Jure Leskovec

Received: 31 January 2013 / Revised: 8 September 2013 / Accepted: 16 September 2013 / Published online: 4 October 2013 © Springer-Verlag London 2013

Abstract Nodes in real-world networks organize into densely linked communities where edges appear with high concentration among the members of the community. Identifying such communities of nodes has proven to be a challenging task due to a plethora of definitions of network communities, intractability of methods for detecting them, and the issues with evaluation which stem from the lack of a reliable gold-standard ground-truth. In this paper, we distinguish between structural and functional definitions of network communities. Structural definitions of communities are based on connectivity patterns, like the density of connections between the community members, while functional definitions are based on (often unobserved) common function or role of the community members in the network. We argue that the goal of network community detection is to extract functional communities based on the *connectivity structure* of the nodes in the network. We then identify networks with explicitly labeled functional communities to which we refer as ground-truth communities. In particular, we study a set of 230 large real-world social, collaboration, and information networks where nodes explicitly state their community memberships. For example, in social networks, nodes explicitly join various interest-based social groups. We use such social groups to define a reliable and robust notion of ground-truth communities. We then propose a methodology, which allows us to compare and quantitatively evaluate how different structural definitions of communities correspond to ground-truth functional communities. We study 13 commonly used structural definitions of communities and examine their sensitivity, robustness and performance in identifying the ground-truth. We show that the 13 structural definitions are heavily correlated and naturally group into four classes. We find that two of these definitions, Conductance and Triad participation ratio, consistently give the best performance in identifying ground-truth communities. We also investigate a task of detecting communities given a single seed node. We extend the local spectral clustering algorithm into a heuristic parameter-free community detection method that easily scales to networks with more than 100 million nodes. The proposed method achieves 30 % relative improvement over current local clustering methods.

J. Yang (⊠) · J. Leskovec Stanford University, Stanford, CA, USA e-mail: crucis@stanford.edu

**Keywords** Network communities · Ground-truth communities · Community detection · Modularity · Community scoring function

# **1** Introduction

Networks are a natural way to represent social [23], biological [26], technological [18], and information [9] systems. Nodes in such networks organize into densely linked groups that are commonly referred to as *network communities*, clusters, or modules [12]. There are many reasons why nodes in networks organize into densely linked clusters. For example, society is organized into social groups, families, villages, and associations [8,14]. On the World Wide Web, topically related pages link more densely among themselves [9]. And, in metabolic networks, densely linked clusters of nodes are related to functional units, such as pathways and cycles [26].

In community detection, one aims to identify sets of nodes that correspond to communities. One way to formalize the process of community detection is to think of a *scoring function* that quantifies how much the connectivity pattern of a given set of nodes resembles the connectivity structure of a network community. Most scoring functions, like Modularity [25] and Conductance [31], are based on the intuition that sets of nodes that have many connections between its members correspond to communities. Once the scoring function is defined then one applies a procedure to find sets of nodes with high score. Such sets of nodes are then extracted and referred to as *network communities*.

Identifying such communities in networks [7, 10, 16, 29, 36] has proven to be a challenging task [11, 19, 20] due to several reasons: There exist a plethora of definitions, scoring functions, and methods for extracting network communities [6, 27]; even if we would agree on a single common structural definition (i.e., a single scoring function), the algorithmic formalizations of community detection lead to NP-hard problems [29]; And the lack of reliable ground-truth makes the evaluation of extracted communities and comparison of algorithms extremely difficult.

Currently, the performance of community detection methods is often evaluated by manual inspection. For each detected community, an effort is made to interpret it as a 'real' community by identifying a common property or external attribute shared by all the members of the community. For example, when examining communities in a scientific collaboration network, we might by manual inspection discover that many of detected communities correspond to groups of scientists working in common areas of science [25]. Such anecdotal evaluation procedures require extensive manual effort, are non-comprehensive, and limited to small networks.

A possible solution to this problem would be to find a reliable definition of explicitly labeled gold-standard ground-truth communities. Using such ground-truth communities would allow for *quantitative* and *large-scale* evaluation and comparison of network community detection methods. Such ability would enable the field to move beyond the current standard of anecdotal evaluation of communities to a comprehensive evaluation of community detection methods based on their performance to extract the ground-truth. Furthermore, it would allow for the development of new community detection methods and improve the understanding of how communities manifest themselves in networks.

#### 1.1 Present work: structure and function

In this paper, we define a robust notion of ground-truth communities. We achieve this by distinguishing between *structural* and *functional* definitions of communities. We argue that

the goal of network community detection is to extract *functional* communities based on the *connectivity structure* of the nodes in the network. We identify networks with explicitly labeled functional communities and then present a methodology that allows us to evaluate different structural definitions of communities.

Generally, after some community detection algorithm identifies communities based on the network structure, the essential next step is to interpret the communities by identifying a common external property or a function that the members of a given community share and around which the community organizes [8]. For example, given a protein–protein interaction network of a cell, one first identifies communities based on the structure of the network and then examines that these communities correspond to real functional units of a cell. Thus, the goal of community detection is to identify sets of nodes with a common (often external/latent/unobserved) *function* based on the connectivity *structure* of the network. In this context, a *common function* can be a common role, affiliation, or attribute [14]. In our protein interaction network example above, such common function of nodes would be 'belonging to the same functional unit.' Or, in a social network, common function would be 'belonging to the same social circle.'

Thus, community detection methods identify communities based on the network *structure*, while the detected communities are then evaluated based on their *function*. Thus, we distinguish between *structural* and *functional* definitions of communities. Structural definitions are based on the structure of the connectivity between a set of nodes (e.g., a set of nodes with high Modularity score [25]). On the other hand, functional definitions of network communities are based on common function or role that the community members share (e.g., proteins of the same functional unit). Generally, the basic premise behind the network community detection is that functional communities have distinct structural patterns, and thus, one may be able to identify them based on the network structure.

1.2 Present work: networks with ground-truth communities

Our goal here is to obtain high-quality labels of ground-truth communities so that we can then devise a methodology to compare and evaluate various structural definitions of network communities.

While explicitly labeled structural communities are nearly impossible to obtain, our main insight here is that there exist networks where functional communities are *explicitly* declared in the data. Thus, we use sets of nodes with a common function to define ground-truth communities.

We gathered 230 networks from a number of different domains and research areas where nodes explicitly state their ground-truth functional community memberships. The size of the networks ranges from hundreds of thousand to hundreds of millions of nodes and edges. The networks represent a wide range of edge densities, numbers of explicitly defined communities, as well as sizes and amounts of community overlap.

Our collection consists of social, collaboration, and information networks for each of which we find a robust functional definition of ground-truth. For example, in online social networks (like Orkut, LiveJournal, and Friendster), we consider explicitly defined *interest-based groups* (e.g., fans of pop singer Lady Gaga, students of the same school) as ground-truth functional communities. Nodes in these networks explicitly join such groups that organize around specific topics, interests, and affiliations [8, 14]. We also consider the product co-purchasing network from Amazon where we define ground-truth using hierarchically nested product categories. Here, all members (i.e., products) of the same ground-truth community share a common function or purpose. Last, in the scientific collaboration network of DBLP, we use

publication venues as proxies for ground-truth research communities. Our reasoning here is that in scientific collaboration networks, real communities would correspond to areas of science. Thus, we use journals and conferences as proxies for (heavily overlapping) scientific communities.

#### 1.3 Present work: methodology and findings

The availability of ground-truth allows us to examine how well various structural definitions of network communities correspond to functional communities (i.e., ground-truth communities). A good structural definition of a community would be such that it would correspond to connectivity patterns that correspond to functional communities. Our experiments show a clear connection between functional and structural definitions: We show that functional communities exhibit distinct connectivity patterns. This means that we can evaluate different structural definitions based on their ability to identify ground-truth communities.

We study 13 commonly used structural definitions of communities and examine their quality, sensitivity, and robustness. Each such definition corresponds to a scoring function that scores a given set of nodes how 'community-like' it is, i.e., a scoring function assigns high score to sets of nodes that closely resemble functional communities. By comparing correlations of scores that different structural definitions assign to ground-truth communities, we find that the 13 definitions naturally group into four distinct classes. These classes correspond to definitions that consider: (1) only internal community connectivity, (2) only external connectivity of the nodes to the rest of the network; (3) both internal and external community connectivity, and (4) network modularity.

We then consider an axiomatic approach and define four intuitive properties that communities would ideally have. Intuitively, a 'good' community is cohesive, compact, and internally well connected while being also well separated from the rest of the network. This allows us to characterize which connectivity patterns a given structural definition detects and which ones it misses. Next, we also investigate the robustness of community scoring functions based on four types of randomized perturbation strategies. Overall, evaluation shows that among the scoring functions considered here those that are based on triadic closure [35] and the Conductance score [31] best capture the structure of ground-truth communities.

Last, we also investigate a task of discovering all members of a community given a single member node. We extend the local spectral clustering algorithm [3] into a parameter-free community detection method that scales to networks of hundreds of millions of nodes. Our method recovers ground-truth communities with 30% relative improvement in the F1-score over the current local graph partitioning methods.

To the best of our knowledge, our work is the first to use social and information networks with explicit community memberships to define an evaluation methodology for comparing network community detection methods based on their accuracy on real data. We believe that the present work will bring more rigor to the standard for the evaluation of community detection methods. All our datasets can be downloaded at http://snap.stanford.edu.

The rest of the paper is organized as follows. Section 2 describes the datasets and defines the notions of ground-truth communities in each dataset. Section 3 shows the distribution of the properties of ground-truth communities and the structural characteristics of ground-truth communities. Section 4 describes the structural definitions of communities that we consider in this paper and discusses the relationship among the definitions. In Sect. 5, we evaluate the structural definitions of communities on two aspects. First, we study what connectivity patterns various definitions prefer and which they penalize. Second, we evaluate the robustness of community structure by using a set of randomized community perturbation strategies.

Section 6 considers the problem of identifying ground-truth communities from seed nodes. Section 7 discusses related work. We conclude in Sect. 8.

Last, we also note that a shorter version of this paper appeared at the IEEE International Conference on Data Mining (ICDM) [38].

#### 2 Ground-truth communities

We begin by explaining the intuition behind the definition the ground-truth communities. We distinguish between *structural* and *functional* definitions of communities. A *structural definition* of communities is a set of nodes with a particular *connectivity structure* (e.g., set of nodes with high edge density or set of nodes with high Modularity score). A *functional definition* of communities is a set of nodes with a *common function*, which can be common role, affiliation, or attribute [8, 14].

With these two definitions of communities, community detection process generally follows a two-step procedure: First one discovers communities based on a structural definition. And then one argues that the discovered communities correspond to functional communities. For example, Palla et al. [26] identified structural communities by identifying sets of overlapping *k*-cliques on protein–protein interaction networks. Then, they found that these structurally defined communities of proteins correspond to functional modules of proteins. In this example, communities are extracted based on the structural definition and then evaluated based on the functional definition by arguing that 'belonging to the same functional module' is the common function of nodes. An issue with this approach is that it is ad hoc and that the evaluation of extracted structural communities is manual—each extracted community has to be manually inspected.

Our approach takes the different direction: We first identify large-scale datasets where functional communities are already labeled, and then we evaluate community detection methods based on their ability to extract ground-truth functional communities.

Overall we consider 230 large social, collaboration and information networks, where for each network we have a graph and a set of functionally defined ground-truth communities. Members of these ground-truth communities share a common function, property or purpose. Networks that we study come from a wide range of domains and sizes. Table 1 lists the networks and their properties.

2.1 Ground-truth communities in social networks

First, we consider three online social networks: the LiveJournal blogging community [5], the Friendster online network [23], and the Orkut social network [23]. In these networks, users

Dataset	Ν	Ε	С	S	Α
LiveJournal	4.0M	34.9M	311,782	40.06	3.09
Friendster	117.7M	2,586.1M	1,449,666	26.72	0.32
Orkut	3.0M	117.2M	8,455,253	34.86	95.9
Ning (225 nets)	7.0M	35.5M	137,177	46.89	0.92
Amazon	0.33M	0.92M	49,732	99.86	14.83
DBLP	0.42M	1.34M	2,547	429.79	2.56

Table 1 230 Social, collaboration, and information networks with explicit ground-truth communities

N number of nodes, E number of edges, C number of communities, S average community size, and A community memberships per node. Ning statistics are aggregated over 225 different subnetworks

create explicit functional groups to which other users then join. Such groups serve as organizing principles of nodes in social networks. Groups range from small to very large and are created based on specific topics, interests, hobbies, and geographical regions. For example, LiveJournal categorizes communities into the following types: culture, entertainment, expression, fandom, life/style, life/support, gaming, sports, student life, and technology. There are over 100 communities in LiveJournal with 'Stanford' in their name, and they range from communities based on different classes, student ethnic communities, departments, activity and interest-based groups, varsity teams, etc. Overall, there are over three hundred thousand explicitly defined communities in LiveJournal.

Similarly, users in Friendster as well as in Orkut define topic-based communities that others then join. Both networks have more than a million explicitly defined groups and each user can join to one or more such groups. We consider each group as a ground-truth community.

Last, we have a set of 225 different online social networks [15] that are all hosted by the Ning platform. It is important to note that each Ning network is a separate social network—an independent Web site with a separate user community. For example, the NBA team Dallas Mavericks and diabetes patients network TuDiabetes all use Ning to host their separate online social networks. After joining a specific network, users then create and join groups. For example, in TuDiabetes, Ning network groups form around specific types of diabetes, parenting children with diabetes, different geographical regions, age groups, and similar. Note that these are exactly the properties around which we expect communities to form in a network of diabetes patients. Again, we use such explicitly defined functional groups as ground-truth communities.

As we see in Table 1, ground-truth communities in social networks are quite diverse. For example, communities in Friendster are about twice smaller than communities in Ning or LiveJournal. Communities in Orkut overlap heavily as people are members of many communities at the same time, while for example, in Friendster, most nodes do not belong to any community.

#### 2.2 Ground-truth communities in product networks

The second type of a network we consider is the Amazon product co-purchasing network [18]. The nodes of the network represent products and edges link commonly co-purchased products. Each product (i.e., node) belongs to one or more hierarchically organized product categories, and products from the same category define a group which we view as a ground-truth community. Note that here the definition of ground-truth is somewhat different. In this case, nodes that belong to a common ground-truth community share a common function or purpose.

Ground-truth communities in product networks (Table 1) are larger than in social networks and include around 100 nodes on the average. Given the hierarchical categorization of products, we also note that an average product belongs to 14 categories, i.e., ground-truth communities.

#### 2.3 Ground-truth communities in collaboration networks

Finally, we also consider the DBLP scientific collaboration network [5], where nodes represent authors and edges connect authors that have co-authored a paper. To define ground-truth in this setting, we reason as follows. Functional communities in a scientific domain correspond to people working in common areas and subareas of science. However, note that publication venues serve as good proxies for scientific areas: People publishing in the same conference form a scientific community. Thus, we use publication venues (i.e., conferences, journals) as ground-truth communities, which serve as proxies for highly overlapping scientific communities around which the collaboration network then organizes.

Ground-truth communities in the DBLP network (Table 1) are the largest and moderately overlap with nodes being part of about 2.5 different communities on the average.

To conclude, we note that all our networks and the corresponding ground-truths are complete and publicly available at http://snap.stanford.edu. For each of these networks, we identified a sensible way of defining ground-truth communities that serve as organizational units of these networks. We were careful to define ground-truth communities based on common affiliation, social circle, role, activity, interest, function, or some other property around which networks organize into communities [8, 14]. Even though our networks come from very different domains and have very different motivation for the formation of communities, the results we present here are consistent and robust. Our work is consistent with the premise that is implicit in all network community literature: members of real communities share some (latent/unobserved) property or affiliation that serves as an organizing principle of the nodes and makes them well connected in the network. Here, we use these groups around which communities organize to explicitly define ground-truth. And, as we will later see, the groundtruth communities exhibit connectivity patterns that match our intuition of communities as densely connected sets of nodes.

#### 2.4 Data preprocessing

To represent all networks in a consistent way, we consider each network as an unweighted undirected static graph. Because members of the group may be disconnected in the network, we consider each connected component of the group as a separate ground-truth community. However, we allow ground-truth communities to be nested and to overlap, i.e., nodes can be members of multiple communities at once.

# 3 Characteristics of ground-truth communities

In this section, we examine properties of ground-truth communities. First we study size and overlap distributions of communities and then proceed to examine finer structural properties of ground-truth communities.

# 3.1 Global properties of ground-truth communities

We start by analyzing the distribution of the properties of ground-truth communities. Figure 1 gives the distributions (complementary CDF) of community sizes which are the number of the nodes in the communities. Notice that all distributions are heavily skewed with most ground-truth communities being small, while large communities also exist. For example, largest social communities contain between one and ten thousand people, while product communities can be even larger.

To get a sense of how much communities overlap, we also examine how many communities a node belong to. Figure 2 plots the distribution of the number of community memberships that a node belongs to. Again, we observe heavy tails with most nodes belonging to only a small number of communities and few nodes belonging to many.

We further examine the properties of community overlaps. We focus on characterizing the overlap between pairs of ground-truth communities. We show the distribution of the



**Fig. 1** Ground-truth community size distribution. Complementary cumulative distribution function of the size of ground-truth communities. The size of a ground-truth community denotes the number of nodes belonging to the ground-truth community



**Fig. 2** Node membership distribution. Complementary cumulative distribution function of the node memberships (the number of communities nodes belong to)

absolute overlap sizes (the number of the nodes in the overlap) in Fig. 3. We observe that the distributions follow a power law, as also observed by Palla et al. [26] on detected (rather than ground-truth) communities.

Last, we also study on the relative size of community overlaps. Relative sizes of overlaps are of our interest as they can characterize how ground-truth communities overlap: Do ground-truth communities overlap in a nested structure? Or, do they overlap only for a small fraction of members? We measure the fraction f of the size of the overlap  $A \cap B$  between two



**Fig. 3** Community overlap distribution. Complementary cumulative distribution function of the size of overlaps between pairs of ground-truth communities. The size of an overlap is the number of the nodes that belong to the overlap



**Fig. 4** Relative size of community overlaps. Histogram of the fraction of the relative overlap size. When ground-truth communities *A*, *B* overlap  $A \cap B$ , then the fraction of the relative overlap size is  $|A \cap B|/\min(|A|, |B|)$ , where  $\min(x, y)$  is the smaller of *x* and *y* 

communities A, B to the size of the smaller community,  $\min(|A|, |B|)$  ( $f = |A \cap B|/\min(|A|, |B|)$ ). If the fraction of overlap is close to 1, the network has a nested structure where the smaller community is included by the larger community. On the other hand, f being close to 0 means that most communities are non-overlapping. We plot the histogram of the overlap fraction in Fig. 4. The Amazon network shows high probability at f = 1 because the ground-truth communities form a nested structure by construction. In social networks and

the DBLP network, most overlaps take a small fraction of individual communities, which is reasonable as each community has its own special interests.

3.2 Structural characteristics of ground-truth communities

In this subsection, we show that ground-truth communities that we defined have distinct connectivity properties. We show that our ground-truth communities, which are sets of nodes sharing common functions or properties (i.e., functional communities), also exhibit distinctive structural properties. The experiments confirm the premise that the functional communities exhibit distinct structural connectivity patterns and can thus be discovered based on the network connectivity structure.

We compare structural properties of a ground-truth community  $C_i$  to those of a set of nodes that do not form a ground-truth community with the goal to establish how ground-truth communities structurally differ from non-communities. For each ground-truth community  $C_i$ , we sample a 'non-community'  $\tilde{C}_i$ , a set of nodes outside  $C_i$  to which we compare  $C_i$ . To make our experiments realistic, we add three constraints to  $\tilde{C}_i$ :

- 1.  $\tilde{C}_i$  has the same number of nodes as  $C_i$
- 2.  $\tilde{C}_i$  is connected
- 3. Members of  $\tilde{C}_i$  have the same distribution of shortest path distances as  $C_i$

The last constraint is an approximation for the ideal that we want  $C_i$  and  $\tilde{C}_i$  to have similar 'compactness' or 'connectedness.' To achieve these constraints, we proceed as follows. We take a node  $u \in C_i$  uniformly at random and compute the distance histogram  $H_u(k) = |\{v \in C_i : d(u, v) = k\}|$  that is the number of other member nodes who are k-hop away from u (k = 0, 1, 2, ...). Then, we pick  $\tilde{u} \notin C_i$  from which we generate  $\tilde{C}_i$  by adding  $H_u(k)$  nodes from the k-hop neighbors of u'. For example, if  $H_u(0) = 1$ ,  $H_u(1) = 3$ ,  $H_u(2) = 5$ ,  $\tilde{C}_i$  contains  $\tilde{u}$ , 3 neighbors of  $\tilde{u}$ , and 5 2-hop neighbors of  $\tilde{u}$ . At the same time, we only choose the nodes that are connected to at least one of the other members of  $\tilde{C}_i$  to guarantee the connectedness of  $\tilde{C}_i$ .

We then measure the structural properties of  $C_i$  and  $\tilde{C}_i$ . For a set of nodes S ( $S = C_i$  or  $\tilde{C}_i$ ) that has  $n_S$  member nodes and  $m_S$  edges among its member nodes, we measure the following:

- *Clustering coefficient (CCF)* is the average clustering coefficient between the member nodes of *S* [35].
- Average degree (AvgDeg) is the average number of node degree to other member nodes.  $(2m_S/n_S)$  [27].
- Edge density (Density) is the fraction of pairs of member nodes that have an edge  $(4m_S/(n_S(n_S 1)))$  [27].
- *Cohesiveness* captures the intuition that a good community should be internally well and evenly connected, i.e., it should be relatively hard to split a community into two subcommunities [19]. We capture this intuition by defining cohesiveness as the Conductance of the internal cut:  $\max_{S' \subset S} \phi(S')$  where  $\phi(S')$  is the Conductance of *S'* measured in the induced subgraph by *S*. We will precisely define Conductance later, but intuitively, the more cohesive the community, the more edges have to be cut in order to further split the community and thus the higher the Conductance score of the internal cut.

Table 2 shows the ratio between the average value of  $C_i$  and that of  $\tilde{C}_i$  for the 3 properties. We observe that groups show 18% higher clustering coefficient, 51% higher average degree, 39% higher edge density, and 102% higher cohesiveness than sets of randomly chosen

Dataset	CCF	AvgDeg	Density	Cohesiveness
LiveJournal	1.18	1.79	1.54	2.68
Friendster	1.13	1.82	1.61	1.24
Orkut	1.11	1.47	1.45	1.17
Ning	1.48	1.39	1.20	1.31
Amazon	1.06	1.29	1.25	3.66
DBLP	1.09	1.33	1.30	2.06
Average	1.18	1.51	1.39	2.02

 Table 2
 Comparison between ground-truth communities (functional communities) and sets of randomly chosen connected nodes

Ratio between the community scores of a ground-truth community over those of a set of connected nodes with the same size and the same distance distribution. *CCF* clustering coefficient, *AvgDeg* average degree of a node to other member nodes, *Density* the fraction of pairs of members that have an edge, *Cohesiveness* the highest Conductance among possible internal cuts. Values higher than 1 mean that ground-truth communities achieve higher score than corresponding non-community sets of nodes

connected sets of nodes. This shows that the members of functional communities tend to be more cohesively and densely connected and thus exhibit distinct connectivity patterns.

#### 4 Community scoring functions

In community detection, one aims to identify sets of nodes that correspond to communities. One way to formalize this process is to design a *scoring function* that for a set of nodes outputs a quality score that characterizes how much the connectivity structure of a given set of nodes resembles that of a community. The idea then is that given a community scoring function, one can then find sets of nodes with high score and consider these sets as communities.

In practice, nearly all scoring functions build on the intuition that communities are sets of nodes with many connections between the members and few connections from the members to the rest of the network. There are many possible ways to mathematically formalize this intuition. We gather 13 commonly used scoring functions or, equivalently, 13 structural definitions of network communities. Some scoring functions are well known in the literature, while others are proposed here for the first time.

Given a set of nodes *S*, we consider a function f(S) that characterizes the community quality of a given set of nodes *S*. Let G(V, E) be an undirected graph with n = |V| nodes and m = |E| edges. Let *S* be the set of nodes, where  $n_S$  is the number of nodes in  $S, n_S = |S|$ ;  $m_S$  the number of edges in  $S, m_S = |\{(u, v) \in E : u \in S, v \in S\}|; c_S$ , the number of edges on the boundary of  $S, c_S = |\{(u, v) \in E : u \in S, v \notin S\}|;$  and d(u) is the degree of node u. We consider 13 scoring functions f(S) that capture the notion of quality of a network community *S*. The experiments we will present later reveal that scoring functions naturally group into the following four classes:

# (A) Scoring functions based on internal connectivity:

- Internal density:  $f(S) = \frac{m_S}{n_S(n_S-1)/2}$  is the internal edge density of the node set S [27].
- *Edges inside:*  $f(S) = m_S$  is the number of edges between the members of S [27].
- Average degree:  $f(S) = \frac{2m_S}{n_S}$  is the average internal degree of the members of S [27].
- Fraction over median degree (FOMD):

 $f(S) = \frac{|\{u:u \in S, |\{(u,v):v \in S\}| > d_m\}|}{n_S}$  is the fraction of nodes of S that have internal degree higher than  $d_m$ , where  $d_m$  is the median value of d(u) in V.

Triangle participation ratio (TPR):
 f(S) = <sup>|[u:u∈S,[(v,w):v,w∈S,(u,v)∈E,(u,w)∈E,(v,w)∈E]≠Ø]|</sup>/<sub>ns</sub> is the fraction of nodes in S that belong to a triad.

# (B) Scoring functions based on external connectivity:

- *Expansion* measures the number of edges per node that point outside the cluster:  $f(S) = \frac{c_S}{n_S}$  [27].
- *Cut ratio* is the fraction of existing edges (out of all possible edges) leaving the cluster:  $f(S) = \frac{c_S}{n_S(n-n_S)}$ [10].

# (C) Scoring functions that combine internal and external connectivity:

- Conductance:  $f(S) = \frac{c_S}{2m_S + c_S}$  measures the fraction of total edge volume that points outside the cluster [31].
- Normalized Cut:  $f(S) = \frac{c_S}{2m_S + c_S} + \frac{c_S}{2(m-m_S) + c_S}$  [31].
- Maximum-ODF (Out Degree Fraction):
   f(S) = max<sub>u∈S</sub> |[(u,v)∈E:v∉S]]/d(u) is the maximum fraction of edges of a node in S that point outside S [9].
- Average ODF:  $f(S) = \frac{1}{n_S} \sum_{u \in S} \frac{|[(u,v) \in E: v \notin S]|}{d(u)}$  is the average fraction of edges of nodes in S that point out of S [9].
- Flake ODF:  $f(S) = \frac{|\{u:u \in S, |\{(u,v) \in E: v \in S\}| < d(u)/2\}|}{n_S}$  is the fraction of nodes in S that have fewer edges pointing inside than to the outside of the cluster [9].

# (D) Scoring function based on a network model:

- *Modularity:*  $f(S) = \frac{1}{4}(m_S E(m_S))$  is the difference between  $m_S$ , the number of edges between nodes in *S*, and  $E(m_S)$ , the expected number of such edges in a random graph with identical degree sequence [24].
- 4.1 Experimental result: four classes of scoring functions

We examine relationship of the 13 community scoring functions we introduced. For each of the 10 million ground-truth communities in our networks, we compute a score using each of the 13 scoring functions. We then create a correlation matrix of scoring functions and threshold it. Figure 5 shows connections between scoring functions with correlation  $\geq 0.6$  on the LiveJournal network.

We observe that scores naturally group into four clusters. This means that scoring functions of the same cluster return heavily correlated values and quantify the same aspect of connectivity structure. Overall, none of the scoring functions are negatively correlated, which means that none of them systematically disagree. Interestingly, Modularity is not correlated with any other scoring function (average degree is the most correlated at 0.05 correlation).

We observe very similar results in all other network datasets that we considered in this study.

The experiment demonstrates that even though many different structural definitions of communities have been proposed, these definitions are heavily correlated. Essentially, there are only 4 different structural notions of network communities as revealed by Fig. 5. For brevity, in the rest of the paper, we present results for 6 representative scoring functions



Fig. 5 Correlations of community scoring functions. Two scoring functions are connected by an edge if the values output by scoring functions are correlated with correlation coefficient  $\geq$ 0.6. Notice four distinct classes of scoring functions

(denoted as blue nodes in Fig. 5): 4 from the two large clusters and 2 from the two small clusters.

We also note that here we computed the values of the 13 scores on ground-truth communities. In reality, the aim of community detection is to find sets of nodes that maximize a given scoring function. Exact maximization of these functions is typically NP-hard and leads to its own set of interesting problems. (Refer to [19] for discussion.)

#### **5** Evaluation of community scoring functions

The second main purpose of our paper is to develop an evaluation methodology for network community detection. Based on ground-truth communities, we now aim to compare and evaluate different community scoring functions. We focus on two aspects of community scoring functions: how well the community scoring function can detect communities (Sect. 5.1) and how robust the community scoring function is to noise in network structure as well as node labeling (Sect. 5.2).

## 5.1 Detecting communities

Our goal is to rank different structural definitions of network communities (i.e., community scoring functions) by their ability to detect ground-truth communities. We adopt the following axiomatic approach. First, we define four community 'goodness' metrics that formalize the intuition that 'good communities' are both compact and well connected internally while being relatively well separated from the rest of the network.

The difference between community scoring functions from Sect. 4 and the goodness metrics defined above is that a community scoring function quantifies how community-like a set is, while a goodness metric in an axiomatic way quantifies a desirable property of a community. A set with high goodness metric does not necessarily correspond to a community, but a set with high community score should have a high value on one or more goodness metrics. In other words, the goodness metrics shed light on various (in many cases mutually exclusive) aspects of the network community structure.

Using the notation from Sect. 4, we define four goodness metrics g(S) for a node set S:

- *Separability* captures the intuition that good communities are well separated from the rest of the network [10,31], meaning that they have relatively few edges pointing from set *S* to the rest of the network. Separability measures the ratio between the internal and the external number of edges of *S*:  $g(S) = \frac{m_S}{c_S}$ .
- *Density* builds on intuition that good communities are well connected [10]. It measures the fraction of the edges (out of all possible edges) that appear between the nodes in *S*,  $g(S) = \frac{m_S}{n_S(n_S-1)/2}$ .
- *Cohesiveness* characterizes the internal structure of the community. Intuitively, a good community should be internally well and evenly connected, i.e., it should be relatively hard to split a community into two subcommunities. We characterize this by the Conductance of the internal cut. Formally,  $g(S) = \max_{S' \subset S} \phi(S')$  where  $\phi(S')$  is the Conductance of *S'* measured in the induced subgraph by *S*. Conductance essentially corresponds to the ratio of the edges in *S'* that point outside the set and the edges inside the set *S'*. A good community should have high cohesiveness (high internal Conductance) as it should require deleting many edges before the community would be internally split into disconnected components [19].
- Clustering coefficient is based on the premise that network communities are manifestations
  of locally inhomogeneous distributions of edges, because pairs of nodes with common
  neighbors are more likely to be connected with each other [35].

#### 5.1.1 Experimental setup

We are interested in quantifying how 'good' are the communities chosen by a particular scoring function f(S) by evaluating their goodness metric. We formulate our experiments as follows: For each of 230 networks, we have a set of ground-truth communities  $S_i$ . For each community scoring function f(S), we rank the ground-truth communities by the decreasing score  $f(S_i)$ . We measure the cumulative running average value of the goodness metric g(S) of the top-k ground-truth communities (under the ordering induced by  $f(S_i)$ ).

The intuition for the experiments is the following. A perfect community scoring function would rank the communities in the decreasing order of the goodness metric, and thus, the cumulative running average of the goodness metric would decrease monotonically with k. Whereas if a hypothetical community scoring function would randomly rank the communities, then the cumulative running average would be a constant function of k.

#### 5.1.2 Experimental results

We found qualitatively similar results on all our datasets. In this section, we only present results for the LiveJournal network. Results are representative for all other networks. We point the reader to the appendix of the paper for a complete set of results (Figs. 10 and 11).

Figure 6a shows the results by plotting the cumulative running average of separability score for LiveJournal ground-truth communities ranked by each of the six community scoring functions. Curve 'U' presents the upper bound, i.e., it plots the cumulative running average of separability when ground-truth communities are ordered by decreasing separability. If the scoring function would order communities exactly by their value of the goodness score, then optimal curve 'U' would be achieved.

We observe that Conductance (C) and cut ratio (CR) give near optimal performance, i.e., they nearly perfectly order the ground-truth communities by separability. On the other hand,



**Fig. 6** Cumulative average of goodness metrics for LiveJournal communities ranked by each of the six representative scoring functions. *C* conductance and *T* TPR, with high and monotonically decreasing values perform best

we observe Triad participation ratio (T) and Modularity (M) score ground-truth communities in the inverse order of separability (especially for k < 100), which means that they both prefer densely linked sets of nodes.

Similarly, Fig. 6b, c, and d shows the cumulative running average of community density, cohesiveness, and clustering coefficient. We observe that all scoring functions (except Modularity) rank denser, more cohesive and more clustered ground-truth communities higher. For the density metric, the fraction over median degree (D) score performs best for high values of k followed by Conductance (C) and Flake ODF (F). In terms of cohesiveness and clustering coefficient, the Triad participation ratio (T) score gives by far the best results. In all cases, the only exception is the Modularity, which ranks the communities in nearly reverse order of the goodness metric (the cumulative running average increases as a function of k). We note that these are all well-known issues of Modularity [11], but they get further attenuated when tested on ground-truth communities.

The curves in Fig. 6 illustrate the ability of the scoring functions to rank communities for the LiveJournal communities. To further quantify this, we perform the following experiment. For a given goodness metric g and for each scoring function f, we measure the rank of each scoring function in comparison with other scoring functions at every value of k. For example, in Fig. 6a, the rank at k = 100 of Conductance is 1, Cut ratio 2, Flake ODF 3, FOMD 4, Modularity 5, and TPR 6. For every k, we rank the scores and compute the average rank over

Scoring function	Separability	Density	Cohesiveness	Clustering
Conductance (C)	1.0	3.5	3.4	3.1
Flake ODF (F)	3.9	3.6	3.5	4.3
FOMD (D)	4.9	3.0	2.9	2.9
TPR (T)	4.5	2.3	2.1	1.2
Modularity (M)	4.0	5.5	5.7	3.9
Cut ratio (CR)	2.6	3.1	3.2	5.5

 Table 3
 Average scoring function rank for each goodness metric

Conductance gives the highest separability, while Triad participation ratio (TPR) scores best on the remaining metrics

Best performing scores are bolded

all values of k, which quantifies the ability of the scoring function to identify communities with high goodness metric.

Table 3 shows the average rank for each score and each goodness metric. An average rank of 1 means that a particular score always outperforms other scores, while rank of 6 means that the score gives worst ranking out of all 6 scores. We observe that Conductance (C) performs best in terms of separability but relatively bad in the other three metrics. For density, cohesiveness, and clustering coefficient, Triad participation ratio (T) is the best. Perhaps not surprisingly, Triad participation ratio scores badly on separability of ground-truth communities. Thus, Conductance is able to identify well-separated communities, but performs poorly in identifying dense and cohesive sets of nodes with high clustering coefficient. On the other hand, triad participation ratio gives the worst performance in terms of separability but scores the best for the other three metrics.

We conclude that depending on the network, different structural notions of network communities might be appropriate. When the network contains well-separated non-overlapping communities, Conductance is the best scoring function. When the network contains dense heavily overlapping communities, then the Triad participation ratio defines the most appropriate notion of a community. Further research is needed to identify most appropriate structural definitions of communities for various types of networks and functional communities. For example, in social networks, we have both identity-based and bond-based communities [28], and they may in fact have different structural signatures.

Figures 10 and 11 in the appendix show the results for all the networks, where we find similar trends. Interestingly, in Figs. 10 and 11, we also observe that the average goodness metric of the top k communities remains flat but then quickly degrades. We observe the same pattern in all our datasets. Thus, for the remainder of the paper, we focus our attention on a set of the top 5,000 communities of each network based on the average rank over the 6 scores.

#### 5.2 Robustness to perturbation

In this subsection, we evaluate community scoring functions using a set of perturbation strategies. We develop a set of strategies to generate randomized perturbations of ground-truth communities, which allows us to investigate robustness and sensitivity of community scoring functions. Intuitively, a good community scoring function should be such that it is stable under small perturbations of the ground-truth community but degrades quickly with larger perturbations.

Our reasoning is as follows. We desire a community scoring function that scores well when evaluated on a ground-truth community but scores low when evaluated on a perturbed community. In other words, an ideal community scoring function should give a maximal value when evaluated on the ground-truth community. If we consider a slightly perturbed ground-truth community (i.e., a node set that differs very slightly from the ground-truth community), we would want the score to be nearly as good as the score of the original ground-truth community. This would mean that the scoring function is robust to noise. However, if the ground-truth community is perturbed so much that it resembles a random set of nodes, then a good scoring function should give it a low score.

# 5.2.1 Community perturbation strategies

We proceed by defining a set of community perturbation strategies. To vary the amount of perturbation, each perturbation strategy has a single parameter p that controls the intensity of the perturbation. Given p and a ground-truth community defined by its members S, the community perturbation starts with S and then modifies it (i.e., changes its members) by executing the perturbation strategy p|S| times. We define the following perturbation strategies:

- NODESWAP perturbation is based on the mechanism where the community memberships diffuse from the original community through the network. We achieve this by picking a random edge (u, v) where  $u \in S$  and  $v \notin S$  and then swap the memberships (i.e., remove u from S and add v). Note that NODESWAP preserves the size of S but if v is not connected to the nodes in S, then NODESWAP makes S disconnected.
- RANDOM takes community members and replaces them with random non-members. We pick a random node  $u \in S$  and a random  $v \notin S$  and then swap the memberships. Like NODESWAP, RANDOM maintains the size of S but may disconnect S. Generally, RANDOM will degrade the quality of S faster than NODESWAP, since NODESWAP only affects the 'fringe' of the community.
- EXPAND perturbation grows the membership set *S* by expanding it at the boundary. We pick a random edge (u, v) where  $u \in S$  and  $v \notin S$  and add v to *S*. Adding v to *S* will generally decrease the quality of the community. EXPAND preserves the connectedness of *S* but increases the size of *S*.
- SHRINK removes members from the community boundary. We pick a random edge (u, v) where  $u \in S$ ,  $v \notin S$  and remove u from S. SHRINK will decrease the quality of S and result in a smaller community while preserving connectedness.

For a given S, let h(S, p) denote a perturbed version of the community generated by the perturbation h of intensity p.

We now quantify the difference of the score between the unperturbed ground-truth community and its perturbation. We use the Z-score, which measures in the units of standard deviation how much the scoring function changes as a function of perturbation intensity p. For ground-truth community  $S_i$ , the Z-score Z(f, h, p) of community scoring function funder perturbation strategy h with intensity p is as follows:

$$Z(f,h,p) = \frac{E_i[f(S_i) - f(h(S_i, p))]}{\sqrt{Var_i[f(h(S_i, p))]}},$$

where  $E_i[\cdot]$ ,  $Var_i[\cdot]$  are the mean and the variance over communities  $S_i$ , and  $f(h(S_i, p))$ is the community score of perturbed  $S_i$  under perturbation h with intensity p. To measure  $f(h(S_i, p))$ , we run 20 trials of  $h(S_i, p)$  and compute the average value of f. Z-score is the difference between the average community score of true communities  $f(S_i)$  and the average community scores of perturbed communities  $f(h(S_i, p))$  normalized by the standard deviation of community scores of perturbed communities. Since  $f(h(S_i, p))$  are independent for each *i*,  $E_i[f(h(S_i, p))]$  follows a normal distribution by the Central Limit Theorem. Thus, P(z < Z(f, h, p)) gives the probability that  $E_i[f(h(S_i, p))] > E_i[f(S_i)]$  where *z* is a standard normal random variable. We measure *f* so that lower values mean better communities, i.e., we add a negative sign to TPR, Modularity and FOMD. High *Z*-scores mean that  $E_i[f(S_i)]$  is likely to be smaller than  $E_i[f(h(S_i, p))]$  and that  $S_i$  is better than  $h(S_i, p)$  in terms of *f*.

# 5.2.2 Experimental results

For each of the 6 community scoring functions, we measure Z-score for perturbation intensity p ranging between 0.01 and 0.6. This means that we randomly swap between 1 and 60% of the community members and measure the Z-score for each scoring function. For small p, small Z-scores are desirable since they indicate that the scoring function is robust to noise. For high perturbation intensities p, high Z-scores are preferred because this suggests that the community scoring function is sensitive, i.e., as the community becomes more 'random,' we want the scoring function to significantly increase its value.

Figure 7 shows the Z-scores as a function of perturbation intensity p in the LiveJournal online social network. We plot the Z-score for each of the 6 community scoring functions. As expected, the Z-scores increase with p, which means that as the community gets more



Fig. 7 Z-scores as a function of the perturbation intensity. C conductance and T Triad participation ratio best detect the perturbations of LiveJournal ground-truth communities

perturbed, the value of the score tends to decrease. The faster the increase the more sensitive and thus the better the score. For example, in LiveJournal, under the NODESWAP perturbation

Conductance (C) exhibits the highest Z-score after p > 0.2, and it has the steepest curve. Triad participation ratio (T) also exhibits desirable behavior. On the other hand, Modularity (M) score does not change much as we perturb the ground-truth communities. This means that Modularity is not good at distinguishing true communities from randomized sets of nodes. Figure 12 and 13 in the appendix give the same plot for all other networks. We observe similar results.

Interestingly, Modularity (M) fails to achieve increasing Z-score as a function of perturbation intensity p in all the networks except the Ning networks. We note that the key difference in Ning is the size of networks: Ning networks contain 3,000 nodes in average, whereas the other 5 networks contain at least almost a million nodes. Thus, our results show that, while Modularity works well in distinguishing true communities in small networks, it fails in large networks. Interestingly, this observation is consistent with the theoretical limitation of Modularity known as 'the resolution limit' [11]. Fortunato and Barthélemy [11] proved that Modularity is unable to distinguish communities when the network is too 'large' compared to community sizes, and this explains the failure of Modularity for 5 large networks other than Ning. Since the size of communities is generally less than 100 regardless of the network size (Table 1 and reference [20]), most communities are too 'small' for Modularity in these large-scale networks.

# 5.2.3 Sensitivity of community scoring functions

We also quantify the sensitivity of community scoring functions by computing the increase of the Z-score between small (p = 0.05) and large perturbations (p = 0.2). As noted above, we prefer a community scoring function with fast increase of the Z-score as the community perturbation intensity increases. Table 4 displays the difference of the Z-score between a large and a small perturbation: Z(f, h, 0.2) - Z(f, h, 0.05). We compute the average increment across all the 230 networks. A high value of increment means that the score is both robust and sensitive. The score is robust because, at small perturbation (p = 0.05), it maintains low Z-value, while at large perturbation (p = 0.2), it has high Z-value and thus the overall Z-score increment is high.

Conductance is the most robust score under NODESWAP and SHRINK. The Triad participation ratio (T) is the most robust under RANDOM and EXPAND. In both cases, Conductance follows Triad participation ratio closely. We note that the clique percolation method (CPM) [26], which is the state-of-the-art overlapping community detection method, implicitly optimizes

Scoring function	NodeSwap	Random	Expand	Shrink
Conductance (C)	1.06	1.59	0.50	0.45
Flake ODF (F)	0.51	1.15	0.11	0.41
FOMD (D)	0.18	0.57	0.19	0.12
TPR (T)	0.37	1.85	0.74	0.21
Modularity (M)	0.23	0.14	0.03	0.15
Cut ratio (CR)	0.53	0.83	0.13	0.43

 Table 4
 Average absolute increment of the Z-score between small and large community perturbations

Best performing scores are bolded

the Triad participation ratio. CPM will add a node to a community only if the node forms a clique (i.e., a triangle) in the community, and the Triad participation ratio of the detected communities will be 1.

# 5.2.4 Bias of scoring functions

The experiments so far revealed surprisingly large differences in the robustness of different community scoring functions. Interestingly, we also observed that Modularity prefers large communities (score increases under EXPAND). To further investigate the bias of different community scoring functions on the size of the underlying network community, we perform the following experiment. We measure how the Z-score changes as a function of the size of the community while keeping perturbation intensity constant. In particular, we calculate the Z-score of each ground-truth community  $S_i$  at perturbation intensity level p = 0.2, and plot it as a function of the community size  $|S_i|$ . Figure 8 shows the results for the LiveJournal communities. Since p = 0.2 represents relatively large perturbation, high Z-scores are desirable.

We observe that under NODESWAP Conductance is the most robust score and that as the community size increases, robustness of Conductance slightly decreases. For RANDOM and EXPAND, the Triad participation ratio score performs best over the whole range of network community sizes. Generally, best performing scores tend to be more sensitive on small



Fig. 8 Z-scores as a function of the community size

# Algorithm 1 Community detection from a seed node

**Require:** Graph G(V, E), seed node s, scoring function f

- (1) Compute a random walk scores  $r_u$  from seed node *s* using PageRank-Nibble [3].
- (2) Order nodes u by the decreasing value of  $r_u/d(u)$ , where d(u) is the degree of u.
- (3) Compute the community scoring function  $f(S_k)$  of the first k nodes  $f_k = f(S_k = \{u_i | i \le k\})$  for every k.
- (4) Detect local minimal of  $f(S_k)$  and detect one or more communities
- if we want to detect one community then

Find the index  $k^*$  at the first local optima of  $f_k$ . **return**  $\hat{S} = \{v_i | i \le k^*\}$  **else** Find the indices  $k_j^*$  at every local optima of  $f_k$ . **return**  $\hat{S}_j = \{v_i | i \le k_j^*\}$ 

end if

communities. The exception is Modularity. The results for Modularity are consistent with the resolution limit [11] mentioned previously. The Z-score of Modularity is very close to 0 for communities smaller than 100 members, i.e., Modularity cannot distinguish a community and a perturbed community when the community is smaller than around 100 nodes due to resolution limit. However, for large communities, Modularity score shows high Z-scores, except under the EXPAND perturbation, where it favors larger null communities in most networks. We note that we find similar results in other networks (Figs. 14 and 15 in the Appendix).

#### 6 Discovering communities from a seed node

Now we focus on the task of *inferring* communities given a single seed node. We consider two tasks that build on two different viewpoints. The first task is motivated by a community-centric view where we discover all members of community *S* given a single member  $s \in S$ . The second task is motivated by a node-centric view where we want to discover *all* communities that a single node *s* belongs to. This means we discover both the number of communities *s* belongs to and the members of these communities.

#### 6.1 Proposed method

We extend the local spectral clustering algorithm [4,32] into a scalable parameter-free community detection method. The benefits of our method are: First, the method requires no input parameters and is able to automatically detect the number of communities as well as the members of those communities. Second, the computational cost of our method is proportional to the size of the detected community (*not* the size of the network). Thus, our method is scalable to networks with hundreds of millions of nodes.

Our method (Algorithm 1) builds on the findings in Sect. 5: First, we aim to find sets of well-connected nodes around node s. We achieve this by defining a local partitioning method based on random walks starting from a single seed node [3]. In particular, we use the *PageRank-Nibble* random walk method that computes the PageRank vector with error  $< \varepsilon$  in time  $O(1/\varepsilon)$  independent of the network size [4]. The nodes with high PageRank scores from s correspond to the well-connected nodes around s. Moreover, the random walk is 'truncated' as it sets PageRank scores  $r_u$  to 0 for nodes u with  $r_u < \varepsilon$ , for some small



constant  $\varepsilon$  [3]. This way the computational cost is proportional to the size of the detected community and not the size of the network.

After the PageRank-Nibble assigns the proximity scores  $r_u$ , we sort the nodes in decreasing proximity  $r_u$  and proceed to the second step of our algorithm which extends the approach of Spielman and Teng [32]. We evaluate the community score on a set  $S_k$  of all the nodes up to *k*th one (note that by construction  $S_{k-1} \subset S_k$ ). This means that for a chosen community scoring function f, we compute  $f(S_k)$  of the set  $S_k$  that is composed of the top k nodes with the highest random walk score  $r_u$ . The local minima of the function  $f(S_k)$  then correspond to extracted communities.

We detect local minima of  $f(S_k)$  using the following heuristic. For increasing k = 1, 2, ...,we measure  $f(S_k)$ . At some point  $k^*$ ,  $f(S_k)$  will stop decreasing and this  $k^*$  becomes our candidate point for a local minimum. If  $f(S_k)$  keeps increasing after  $k^*$  and eventually becomes higher than  $\alpha f(S_{k^*})$ , we take  $k^*$  as a valid local minimum. However, if  $f(S_k)$  goes down again before it reaches  $\alpha f(S_{k^*})$ , we discard the candidate  $k^*$ . We experimented with several values of  $\alpha$  and found that  $\alpha = 1.2$  gives good results across all the datasets.

For example, Fig. 9 plots  $f(S_k)$  for two community scoring functions f (Conductance) and f' (Triad participation ratio). We identify the local optima (denoted by stars and squares) and use the nodes in the corresponding sets  $S_k$  as the detected communities.

Note that our method can detect multiple communities that the seed node belongs to by identifying different local minima of  $f(S_k)$ . However, we assume that the communities are nested (smaller communities are contained in the larger ones) even though the ground-truth communities may not necessarily follow such a nested structure. Also, note that our method is parameter-free. Our method differs from local graph clustering approaches [3,32] in two aspects. First, instead of sweeping only using Conductance, we consider sweeping using other scoring functions. Second, we find the local optima of the sweep curve instead of the global optimum—this change gives a large improvement over the conventional local spectral clustering approaches [3,32].

# 6.2 Detecting a community from a single member

We first consider the task where we aim to reconstruct a single ground-truth community *S* based on one member node *s*. For each community *S*, we pick a random member node *s* as a seed node and compare the community we detect from *s* with the ground-truth community *S*. Starting from node *s*, we generate a sweep curve  $f(S_k)$ . Let  $k^*$  be the value of *k* where

F1-score	С	F	D	Т	М	CR	LC	CPM
LJ	0.64	0.64	0.62	0.57	0.15	0.61	0.54	0.43
FS	0.23	0.22	0.24	0.25	0.24	0.18	0.13	0.14
Orkut	0.21	0.19	0.19	0.18	0.20	0.09	0.20	0.13
Ning	0.24	0.19	0.10	0.19	0.08	0.19	0.17	0.11
Amazon	0.87	0.75	0.73	0.79	0.06	0.85	0.74	0.85
DBLP	0.61	0.61	0.65	0.66	0.04	0.61	0.46	0.53
Avg. F1	0.46	0.43	0.42	0.44	0.13	0.42	0.37	0.36
Avg. prec	0.50	0.53	0.52	0.55	0.13	0.53	0.49	0.38
Avg. rec	0.60	0.47	0.51	0.47	0.71	0.49	0.65	0.69

 Table 5
 Performance of our 6 methods and 2 baselines (LC, CPM) at detecting communities from a seed node

Best performing scores are bolded

 $f(S_k)$  achieves the first local minima. We then use the set  $S_{k^*}$  as the detected community. Now, given the ground-truth community *S* and the detected community  $S_{k^*}$ , we evaluate how well-detected  $S_{k^*}$  corresponds to ground-truth *S*. For this purpose, we compute precision, recall, and the F1-score between  $S_{k^*}$  and *S*. For all metrics, the score of 1 means that detected  $S_{k^*}$  perfectly matches ground-truth *S*. We consider 6 community scoring functions  $f(\cdot)$  and compare the performance of our method to two standard community detection methods: local spectral clustering (LC) [3] and the 3-clique clique percolation method (CPM) [26].

Table 5 shows the performance of the proposed method for each scoring function and for the two baselines. First 5 rows show the F1-score for each of the datasets, and the last 3 rows show the average F1-score, precision and recall over all the datasets. We observe that the Conductance (C) gives the best average F1-score and outperforms all other scores on LiveJournal (LJ), Orkut, Amazon, and Ning. For Friendster (FS) and DBLP, the Triad participation ratio (T) performs best. This agrees with our intuition that for networks, like LiveJournal, that have fewer community overlaps scoring functions that focus on good separability perform well. In networks where nodes belong to multiple communities (like DBLP where authors publish at multiple venues), the Triad participation ratio (T) performs best. We also note that the average F1-score of Conductance is 0.46, while the baselines CPM and LC achieve F1-score of only 0.36 and 0.37, respectively. Note this is 10% absolute and 30% relative improvement over the baselines.

Last, we observe that some methods detect larger communities than necessary (higher recall, lower precision). Modularity (M) most severely overestimates community size. Conductance (C) and both baselines (CR and CPM) exhibit similar behavior but to a lesser extent. On the contrary, Flake ODF (F), Fraction over median (D), Triad participation ratio (T), and Cut ratio (CR) tend to underestimate the community size (higher precision than recall).

#### 6.3 Detecting all communities that a seed node belongs to

We also explore the second task where we want to detect *all* the communities to which a given seed node *s* belongs. In this task, we are given a node *s* that is a member of multiple communities, but we do not know which and how many communities *s* belongs to. We detect multiple communities by detecting *all* the local minima (and not just the first one) of the sweep curve. This way our method detects both the number and the members of communities.

Network	g (Number of communities)							
	1	2	3	4	≥5	All nodes		
LJ	0.52	0.59	0.52	0.42	0.38	0.53		
FS	0.13	0.10	0.08	0.05	0.02	0.13		
Orkut	0.21	0.17	0.13	0.11	0.10	0.20		
Ning (225 nets)	0.11	0.09	0.07	0.06	0.05	0.11		
Amazon	0.59	0.73	0.69	0.66	0.55	0.61		
DBLP	0.34	0.24	0.20	0.21	0.16	0.33		

Table 6 Average F-score between detected communities and the ground-truth communities to which a seed node belongs to, when the seed node belongs to g different communities

For each dataset, we sample a node s, detect communities  $\hat{S}_j$ , and compare them to the ground-truth communities  $S_i$  that node s belongs to. To measure correspondence between the true and the detected communities, we match ground-truth communities to detected communities by the Hungarian matching method [17]. We then compute the average F1-score over the matched pairs. We use Conductance as the community scoring function and report results in Table 6.

Note that this task is harder than the previous one as here we aim to discover multiple communities simultaneously. Whereas the previous task evaluated our method for each groundtruth community, here, we first sample node *s* and then search for the communities  $S_i$  that *s* belongs to. Therefore, larger ground-truth communities will be included in  $S_i$  more often. Since larger ground-truth communities are less well separated [20], this makes the task harder.

Table 6 reports the average F1-score as a function of the number of communities g that the seed node s belongs to. Given that this is a harder task, we observe overall lower values of the F-score. Moreover, we also expect the performance to decrease as node s belongs to more and more communities. In fact, we observe that the performance degrades with increasing the number of communities g. Interestingly, in LiveJournal and Amazon, it appears to be easier to detect communities of nodes that belong to 2 communities than to detect a community of a node that belongs to only a single community. This is due to the fact that single community nodes reside on the border of the community, and consequently, Conductance produces communities that are too small [20].

# 7 Related work

Generally, there are two approaches toward understanding the characteristics of network community structure and the community scoring functions (i.e., objective functions for community detection).

First way is theoretical analysis, which has been performed for a few most widely used functions such as Modularity [11] and Conductance [3,13]. For example, Gleich and Seshadhri [13] mathematically proved the existence of node sets with high conductance in networks with high clustering coefficient. Fortunato and Barthelemy [11] showed that Modularity may not detect communities that have too few edges compared to the total number of edges in the network. More general theoretical analysis includes the work of Meilă [22], which studied the axiomatic criterion for community scores such as stability [34]. Although these theoretical attempts provide rigorous results, they cover only a few different scoring functions.

To consider a broader range of community scoring functions, therefore, we take a different way, an empirical analysis, which allows us to evaluate any community scoring function.

Empirical approach analyzes how community scoring functions behave on real-world networks. One recent example is by Leskovec et al. [19], which evaluated a wide range of community scoring functions on large-scale real-world networks. However, there are two crucial differences in our work here. First, Leskovec et al. [19] used *detected* communities by the local spectral method [3] for the evaluation. Using communities detected by a specific community detection method would introduce a bias introduced by the detection method [1]. In this paper, our evaluation is free from such bias as we adopt ground-truth communities, which are explicitly declared by individual nodes. Second, Leskovec et al. [19] provides *qualitative* evaluation by showing the Network Community Profile plot [20] for each community scoring function. Here, we aim to quantify the robustness and sensitivity of community scoring functions to compare which scoring functions are better than others. We also note that empirical approach can be done with synthetic benchmarks such as in Shi et al. [30]; however, using synthetic benchmarks generates further biases due to natural deficiencies of synthetically generate networks.

Another related line of research is data-driven analysis of community detection algorithms. For example, Ahn et al. [2] employed community quality metrics based on the purity of node attributes to evaluate the performance of community detection methods. Abrahao et al. [1] showed that the communities detected by different communities exhibit fundamentally different structural properties. In this paper, however, we use data-driven evaluation metric to assess the performance of community detection on ground-truth communities. Our quality metrics directly focus on the correspondence between the detected and the ground-truth communities (as in Lin et al. [21], Sun et al. [33]).

# 8 Conclusion

The lack of reliable ground-truth gold-standard communities has made network community detection a very challenging task. In this paper, we studied a set of 230 different large social, collaboration, and information networks in which we defined the notion of ground-truth communities by nodes *explicitly* stating their group memberships.

We developed an evaluation methodology for comparing network community detection algorithms based on their accuracy on real data and compared different definitions of network communities and examined their robustness. Our results demonstrate large differences in behavior of community scoring functions. Last, we also studied the problem of community detection from a single seed node. We examined class of scalable parameter-free community detection methods based on Random Walks and found that our methods reliably detect ground-truth communities.

The availability of ground-truth communities allows for a range of interesting future directions. For example, further examining the connectivity structure of ground-truth communities could lead to novel community detection methods [37,39,40]. Overall, we believe that the present work will bring more rigor to the evaluation of network community detection, and the datasets publicly released as a part of this work will benefit the research community.

Acknowledgments This research has been supported in part by NSF IIS-1016909, CNS-1010921, CAREER IIS-1149837, IIS- 1159679, ARO MURI, DARPA XDATA, DARPA GRAPHS, ARL AHPCRC, Okawa Foundation, Docomo, Boeing, Allyes, Volkswagen, Intel, Alfred P. Sloan Fellowship, and the Microsoft Faculty Fellowship.

# A Appendix

See Figs. 10, 11, 12, 13, 14, 15.



Fig. 10 Average metrics of *top* k communities by 6 scores. C conductance, T TPR, M modularity, F flake ODF, D FOMD, CR cut ratio



Fig. 11 Average metrics of *top* k communities by 6 scores. C conductance, T TPR, M modularity, F flake ODF, D FOMD, CR cut ratio



**Fig. 12** *Z*-score of 6 scores versus the perturbation intensity for each null model. *C* conductance, *T* TPR, *M* modularity, *F* flake ODF, *D* FOMD, *CR* cut ratio



**Fig. 13** Z-score of 6 scores versus the perturbation intensity for each null model. C conductance, T TPR, M modularity, F flake ODF, D FOMD, CR cut ratio



Fig. 14 Z-score of 6 scores versus the community size for each null model. C conductance, T TPR, M modularity, F flake ODF, D FOMD, CR cut ratio



**Fig. 15** *Z*-score of 6 scores versus the community size for each null model. *C* conductance, *T* TPR, *M* modularity, *F* flake ODF, *D* FOMD, *CR* cut ratio

# References

- Abrahao BD, Soundarajan S, Hopcroft JE, Kleinberg R (2012) On the separability of structural classes of communities. In KDD '12: proceedings of the 18th ACM SIGKDD international conference on knowledge discovery and data mining, pp 624–632
- Ahn Y-Y, Bagrow JP, Lehmann S (2010) Link communities reveal multi-scale complexity in networks. Nature 466:761–764
- Andersen R, Chung F, Lang K (2006) Local graph partitioning using PageRank vectors. In FOCS '06: proceedings of the 47th annual IEEE symposium on foundations of computer science, pp 475–486
- Andersen R, Lang K (2006) Communities from seed sets. In: WWW '06 proceedings of the 15th international conference on, World Wide Web, pp 223–232

- Backstrom L, Huttenlocher D, Kleinberg J, Lan X (2006) Group formation in large social networks: membership, growth and evolution. In KDD '06: proceedings of the 12th ACM SIGKDD international conference on knowledge discovery and data mining, pp 44–54
- Danon L, Duch J, Diaz-Guilera A, Arenas A (2005) Comparing community structure identification. J Stat Mech Theory Exp 29(09):P09008
- Dhillon I, Guan Y, Kulis B (2007) Weighted graph cuts without eigenvectors: a multilevel approach. IEEE Trans Pattern Anal Mach Intell 29(11):1944–1957
- 8. Feld SL (1981) The focused organization of social ties. Am J Sociol 86(5):1015-1035
- Flake G, Lawrence S, Giles C (2000) Efficient identification of web communities. In KDD '00: proceedings of the 6th ACM SIGKDD international conference on knowledge discovery and data mining, pp 150–160
- 10. Fortunato S (2010) Community detection in graphs. Phys Rep 486(3-5):75-174
- Fortunato S, Barthélemy M (2007) Resolution limit in community detection. Proc Nat Acad Sci USA 104(1):36–41
- Girvan M, Newman M (2002) Community structure in social and biological networks. Proc Nat Acad Sci USA 99(12):7821–7826
- Gleich DF, Seshadhri C (2012) Neighborhoods are good communities. In KDD '12: proceedings of the 18th ACM SIGKDD international conference on knowledge discovery and data mining, pp 597–605
- 14. Granovetter MS (1973) The strength of weak ties. Am J Sociol 78:1360-1380
- Kairam S, Wang D, Leskovec J (2012) The life and death of online groups: predicting group growth and longevity. In WSDM '12: ACM international conference on web search and data mining
- Karypis G, Kumar V (1998) A fast and high quality multilevel scheme for partitioning irregular graphs. SIAM J Sci Comput 20:359–392
- 17. Kuhn HW (1955) The Hungarian method for the assignment problem. Naval Res Logist Q 2:83–97
- 18. Leskovec J, Adamic L, Huberman B (2007) The dynamics of viral marketing. ACM Trans Web 1(1):5
- 19. Leskovec J, Lang K, Mahoney M (2010) Empirical comparison of algorithms for network community detection. In WWW '10: proceedings of the 19th international conference on World Wide Web
- Leskovec J, Lang KJ, Dasgupta A, Mahoney MW (2009) Community structure in large networks: natural cluster sizes and the absence of large well-defined clusters. Internet Math 6(1):29–123
- Lin W, Kong X, Yu PS, Wu Q, Jia Y, Li C (2012) Community detection in incomplete information networks. In WWW '12: proceedings of the 21st international conference on, World Wide Web, pp 341–350
- Meilă M (2005) Comparing clusterings: an axiomatic view. In ICML '05: proceedings of the 22nd international conference on machine learning. New York, NY, USA, pp 577–584
- Mislove A, Marcon M, Gummadi KP, Druschel P, Bhattacharjee B (2007) Measurement and analysis of online social networks. In IMC '07: proceedings of the 7th ACM SIGCOMM conference on internet, measurement, pp 29–42
- Newman M (2006) Modularity and community structure in networks. Proc Nat Acad Sci USA 103(23):8577–8582
- Newman M, Girvan M (2004) Finding and evaluating community structure in networks. Phys Rev E 69:026113
- Palla G, Derényi I, Farkas I, Vicsek T (2005) Uncovering the overlapping community structure of complex networks in nature and society. Nature 435(7043):814–818
- Radicchi F, Castellano C, Cecconi F, Loreto V, Parisi D (2004) Defining and identifying communities in networks. Proc Nat Acad Sci USA 101(9):2658–2663
- Ren Y, Kraut R, Kiesler S (2007) Applying common identity and bond theory to design of online communities. Organ Stud 28(3):377–408
- 29. Schaeffer S (2007) Graph clustering. Comp Sci Rev 1(1):27-64
- Shi C, Yu PS, Cai Y, Yan Z, Wu B (2011) On selection of objective functions in multi-objective community detection. In CIKM '11: proceedings of the 20th ACM international conference on information and, knowledge management, pp 2301–2304
- Shi J, Malik J (2000) Normalized cuts and image segmentation. IEEE Trans Pattern Anal Mach Intell 22(8):888–905
- Spielman D, Teng S-H (2004) Nearly-linear time algorithms for graph partitioning, graph sparsification, and solving linear systems. In STOC '04: proceedings of the 36th annual ACM symposium on theory of computing, pp 81–90
- 33. Sun Y, Yu Y, Han J (2009) Ranking-based clustering of heterogeneous information networks with star network schema. In KDD '09: proceedings of the 15th ACM SIGKDD international conference on knowledge discovery and data mining, pp 797–806
- 34. von Luxburg U (2010) Clustering stability: an overview. Found Trends Mach Learn 2(3):235–274

- 35. Watts D, Strogatz S (1998) Collective dynamics of small-world networks. Nature 393:440-442
- Xie J, Kelley S, Szymanski BK (2013) Overlapping community detection in networks: the state of the art and comparative study. ACM Comput Surv 45(4). Art no 43
- Yang J, Leskovec J (2012) Community-affiliation graph model for overlapping network community detection. In ICDM '12: proceedings of the 2012 IEEE international conference on data mining, pp 1170–1175
- Yang J, Leskovec J (2012) Defining and evaluating network communities based on ground-truth. In ICDM '12: proceedings of the 2012 IEEE international conference on data mining, pp 745–754
- Yang J, Leskovec J (2013) Overlapping community detection at scale: a non-negative factorization approach. In WSDM '13: proceedings of the sixth ACM international conference on web search and data mining, pp 587–596
- Yang J, Leskovec J (2013) Structure and overlaps of communities in networks. ACM Trans Intell Syst Technol (to appear)

# **Author Biographies**



Jaewon Yang is a Ph.D. candidate in Electrical Engineering at Stanford University. His research interests include network community detection, non-negative matrix factorization, and the analysis of timeevolving event sequences. He has received the best application paper award from ICDM '10. His Ph.D. study was partially funded by Samsung Scholarship. Jaewon received his bachelor's degree in Electrical Engineering from Seoul National University, South Korea, and master's degree in Statistics from Stanford University.



Jure Leskovec is assistant professor of Computer Science at Stanford University. His research focuses on mining large social and information networks. Problems he investigates are motivated by large scale data, the Web and on-line media. This research has won several awards including best paper awards at KDD, WSDM, ICDM, WWW, ACM KDD dissertation award, Microsoft Research Faculty Fellowship, as well as Alfred P. Sloan Fellowship. Jure received his bachelor's degree in computer science from University of Ljubljana, Slovenia, Ph.D. in machine learning from the Carnegie Mellon University and postdoctoral training at Cornell University.